Author’s response to reviews

Title: MiR-146a G/C rs2910164 variations in South African Indian and Caucasian patients with psoriatic arthritis

Authors:
Ajesh Maharaj (Maharaja30@ukzn.ac.za)
Pragalathan Naidoo (pragalathan.naidoo@gmail.com)
Terisha Ghazi (terishaghazi@gmail.com)
Naeem Abdul (naeemsheik11@gmail.com)
Shanel Dhani (dhanishanel@gmail.com)
Taskeen Docrat (taskeendocrat23@gmail.com)
Prithiksha Ramkaran (prithiksharamkaran@gmail.com)
Paul Tak (p.p.tak@amc.uva.nl)
Niek de Vries (niek.devries@amc.uva.nl)
Anil Chuturgoon (chutur@ukzn.ac.za)

Version: 3 Date: 12 Mar 2018

Author’s response to reviews:

Response Letter

Thank you for the positive feedback. Detailed below is the response to each query:

1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Done. Title Page: Page 1; lines 15-23

2. Please change the Introduction heading to Background.
3. Please change the Materials and Methods heading to Methods.

Done. Methods section: Page 4; line 103

4. We would also like to ask for you to provide more justification for the contributions of NSA, SD, TFD, TG, PR, PPT, NDV, as currently they do not automatically qualify for authorship. Contribution to one aspect of the study, alone, does not usually justify authorship.

Done. Author’s contribution section: P10; lines 315-318

5. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Done. Author’s contribution section: P10; lines 317-318

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Clean copy uploaded.